Author's response to reviews

Title: Validity of the ReproQ; a questionnaire measuring the World Health Organization concept of Responsiveness in Perinatal Care

Authors:

Jacoba van der Kooy (j.vanderkooy@erasusmc.nl)
Nicole B Valentine (n.valentine@who.org)
Erwin Birnie (birnie@bmg.eur.nl)
Marijana Vujkovic (marijana.vujkovic@gmail.com)
Johanna P de Graaf (j.degraaf@erasusmc.nl)
Semiha Denktas (s.denktas@erasusmc.nl)
Eric A.P. Steegers (e.a.p.steegers@erasusmc.nl)
Gouke J. Bonsel (g.bonsel@erasusmc.nl)

Version: 2 Date: 9 April 2014

Author's response to reviews:

Dear editor,

We revised our manuscript according to the ... given.
We added line numbers and removed page breaks.
- Background
- Methods
- Results
- Discussion (Results and Discussion may be combined)
- Conclusions
- List of abbreviations used (if any)

We added an acknowledgement section.